Author’s response to reviews

Title: The impact of early diagnosis on the prognosis of extranodal NK/T-cell lymphoma with massive lung involvement: A case report

Authors:

Tomohiro Yabushita (yabishita527@gmail.com)
Satoshi Yoshioka (satoseay@kcho.jp)
Takeru Furumiya (tfurumiya@ab.auone-net.jp)
Momoko Nakamura (momokonakamura2014@gmail.com)
Daisuke Yamashita (daisuke_yamashita@kcho.jp)
Yukihiro Imai (yukiimai@kcho.jp)
Takayuki Ishikawa (ishikawa@kcho.jp)

Version: 1 Date: 23 Dec 2017

Author’s response to reviews:

December 23, 2017
Re: PULM-D-17-00575

Dear Marta Gritti, BMC Pulmonary Medicine

We are grateful to you for the comments on our manuscript entitled "The impact of early diagnosis on the prognosis of extranodal NK/T-cell lymphoma with massive lung involvement: A case report". We have revised the text and CARE checklist as appropriate.

1. Thank you for providing a CARE checklist.
Please complete the checklist in full by inserting the page number/paragraph and section of your manuscript which reports the information that meets the criteria of the checklist. For example “Methods, paragraph 2”. If a criterion is not applicable for your particular manuscript/study, we can accept “N/A”. Please note that checklists completed incorrectly will be returned for revision as we cannot progress your manuscript to peer review until the checklist has been completed.

Thank you very much for the comment. We’ve completed the CARE checklist by inserting the page number and lines which describes the information that meets the criteria of the CARE checklist.

2. Please amend the headings of the abstract (Background; Methods; Results; Conclusions) and/or of the manuscript (Background; Methods; Results; Discussion; Conclusions) in accordance to BMC Pulmonary Medicine formatting.

Thank you very much for the comment. We’ve changed the headings of the abstract and of the manuscript in accordance with BMC Pulmonary Medicine formatting for case reports.

3. Please amend your Consent for Publication statement into: 'Written informed consent to publish this case report has been obtained from this patient'.

Thank you very much for the comment. We’ve changed the description of Publication statement into 'Written informed consent to publish this case report has been obtained from this patient'.

4. Please move the Tables and the Figure Legends after the References.

Thank you very much for the comment. As you pointed out, we’ve moved the Tables and the Figure Legends after the References.
We would appreciate it if you should contact us when we require further correction. We trust that our manuscript is now suitable for publication in BMC Pulmonary Medicine.

Sincerely,

Thank you for your consideration. We look forward to hearing from you.

Satoshi Yoshioka, M.D., Ph.D.
Chief Physician
Department of Hematology
Kobe City Medical Center General Hospital,
2-1-1, Minatojimaminamimachi, Chuo-ku,
Kobe, 650-0047, Japan
TEL: +81-78-302-4321
FAX: +81-78-302-7537
E-mail: satoseay@kcho.jp